
Daily Status June 15
Weather: hot with clear skies and light variable winds, picking up to a westerly breeze in the afternoon.  An extra sounding was launched last evening in 
support of TLS ops, plus an extra sounding this morning to aid the radiometer intercomparison at the Orange Grove site.  Surface met at sounding site still 
down. Todays soundings at 0Z, 6Z, 9Z, 12Z, and 18Z with the last sounding being in a Perdigao group photo.

The sodar went down again sometime this morning.  Was reset and running again around 14 UTC (see sodar blog entry comments).

The final day of the campaign!
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